GES 235, Spring 1997

Homework #2

Due Thursday , May 1, 1997

1. Revisit Problem 1.1 on p. 34 of Groundwater in Geological Processes. Using the interactive ground water modeling web page, develop a topography-driven flow system that could explain the data. You should provide enough information so that your simulation can be qualitatively reproduced by others.

2. (a) Confirm that eq. (7-13) of Lecture Notes 7 is actually the solution to the ground-water flow problem analyzed by Domenico and Palciauskas (1973). (b) Similarily, confirm that eq. (7-19) is the solution to the heat flow problem (after the approximation is made).

3. (a) Problem 4.3 on p. 136 of Groundwater in Geological Processes. (b) Assuming that the model of Domenico and Palciauskas can be applied to the basin, compute the heat flow rate at the recharge end (x = L, z = H) and discharge end (x = 0, z = H) of the basin as fraction of the basal heat flow rate.